Author's response to reviews

Title: Evidence-based health information from the users' perspective - a qualitative analysis

Authors:

Irene Hirschberg (hirschberg.irene@mh-hannover.de)
Gabriele Seidel (seidel.gabriele@mh-hannover.de)
Daniel Strech (strech.daniel@mh-hannover.de)
Hilda Bastian (hilda.bastian@nih.gov)
Marie-Luise Dierks (dierks.marie-luise@mh-hannover.de)

Version: 6 Date: 7 October 2013

Author's response to reviews: see over
First, we would like to thank the Associate Editor and the Reviewers for their comments and the acceptance of our manuscript. Our resubmitted version addresses both editorial requests; we like to comment in the following:

1. Please adhere to RATS guidelines.

Authors’ response: We confirm that we adhere to the RATS guidelines on qualitative research and have added in the Methods section as following: “Within our study we adhere to the RATS guidelines on qualitative research [24].”

2. Consent statement
Please state in the Methods section whether written informed consent for participation in the study was obtained from participants or, where participants are children, a parent or guardian.

Authors’ response: Prior to participation we have informed all participants in detail on our study (aim, conditions, procedure etc.), granting confidentiality and anonymity. The consent for participation in this user testing of health informational products was obtained orally from all participants themselves. We have integrated the following in the Methods section: “An informed consent for participation in this study was obtained orally from all participants.”

If you have further queries, please let us know. Thank you very much for your support.

Best regards – on behalf of all authors
Irene Hirschberg